From 262157c88d43b1fc42780c2f94aca38c76610777 Mon Sep 17 00:00:00 2001
From: Badlop <badlop@process-one.net>
Date: Wed, 27 Jul 2011 10:20:22 +0200
Subject: Preliminary support for groupname@vhost in Displayed Groups
 (EJAB-506)

---
 doc/guide.tex | 5 +++--
 1 file changed, 3 insertions(+), 2 deletions(-)

(limited to 'doc/guide.tex')

diff --git a/doc/guide.tex b/doc/guide.tex
index 6d12ed2cd..f13c58884 100644
--- a/doc/guide.tex
+++ b/doc/guide.tex
@@ -4118,8 +4118,9 @@ has a unique identification and the following parameters:
   which is only recommended for a small server with just a few hundred users.
   The special member directive \term{@online@}
   represents the online users in the virtual host.
-\item[Displayed groups] A list of groups that will be in the rosters of this
-  group's members.
+\item[Displayed groups]
+  A list of groups that will be in the rosters of this group's members.
+  A group of other vhost can be identified with \term{groupid@vhost}
 \end{description}
 
 Examples:
-- 
cgit v1.2.3

